		EO-1 Weekly Report
	Earth Observing – 1

	November 04, 2015 – November 10, 2015

	DOY
	308
	–
	314

	Mission Day
	5512
	–
	5518


EO-1 SPACECRAFT

Instruments
· Scheduled 119 science Data Collection Events (DCEs)
· Lost 1 scene and parts of 2 other scenes detailed below 
· Performed Instrument Decontamination Cycle 
· Conducted HSI deicing from 313/17:55z to 314/08:45z
· Conducted ALI out-gassing from 313/18:05z to 314/08:55z
· Performed Instrument Calibration 
· Conducted Solar Array Characterization on DOY 314/03:13z
· Conducted HSI Solar Calibration on DOY 314/18:02z

EO-1 MISSION OPERATIONS CENTER
Real-Time
· Maintaining nominal spacecraft state of health 
· Ingesting data to both ITPS machines to make sure both machines match.
· Determining the best way to keep the two systems current either by rsync or shared drive
· Working on upgrading the FD windows XP machines to windows 7
· Ran into a problem with STK license and waiting on SA to fix before being able to continue
· Monitoring star-tracker related mnemonics
· Kalman Filter residual for the X direction is starting to become invalid occasionally. The filter is receiving data from the Star-tracker but failing to pass muster.
· Monitoring ACE LV PC current
· Twice this mnemonic touched yellow high of 3A, still well within the 2.5-3.5A nominal operations and below the 4.94A overcurrent trip
· Monitoring IHSK_T27 ALI Aperture Cover Temperature 
· Temporarily filtered out the yellow low flag
· Trending the data to better understand why the yellow limit is tripping
· Testing Overdraw after update in downtime of FD systems

Mission Planning
· Planning the next set of lunar calibrations for the month of November 
· Working with the USAK04 personnel to bring start passes with that ground station
· Working with ASPEN developers to configure various Lunar Calibration sequences 
· Bringing back Modified Lunar Calibrations

Flight Dynamics 
· Contacted AGI to update the STK PODS tables file. If this file wasn’t updated, the EO-1 MOC wouldn’t have been able to continue performing Orbit Determination. 
· AGI will not update the file any longer
· Received scripts from them that they use to create the tables and are in process to get them working correctly.
· Testing using GPS based tracking data for Orbit Determination
· Verifying rough draft of delta-v planning procedures


Central File Hub
· Modified scripts for incoming FD, MP, and Realtime files to only make directory calls and checks to the destination directory if new files are in. This reduces the amount of ssh calls used by the hubs and cuts down on inner MOC traffic
· Phase one tests are ready for outside entity delivery.
· SGS and WGS tracking data are waiting for a break in WALLOPS activities to begin testing.
· Phase 2 FD scripts have no changes from Phase 1 scripts
· All non-FD scripts are implemented Phase 2 
· Beginning Phase 3 work on deliveries for the central file hub
· Central file hub delivery structure set
· Passwordless ssh set up from FD to eo1out and eo1in to and from eo1out
· Script to put FD products onto file hubs is complete and ready for use
· Script to sort FD products into destination directories is complete and ready for use
· Base delivery script completed
· Designing delivery code architecture
· Reviewing current deliveries for products, emails, and destinations
· Delivery will be completed via FD computers as is the current standard

System Administration
· All MOC linux systems except fluflu2 have had the bash and bind patches applied.
· Awaiting test of patched judo2
· Overdraw update has been completed and is ready for testing
· Participated in weekly security tag up and biweekly FDP meeting.
· [bookmark: _GoBack1]Updated test Waldorf2 to Windows 7 and patched to current level. It is now ready for testing.
· Loaded image of gs428-eo1fot1 on replacement system and patched drivers.
· Defragmented drives on both ITPS systems
· Rebuilding RHEL 6 32 bit virtual machine
· Debugging LiveUpdate issue on eo1aspen1
· Created backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, trending system, regional logs server, and a full backup of the MOC CNE computer.

GROUND AND SPACE NETWORK
Station Downtimes 	
· NAM 0002686 Open: Planned antenna maintenance on TR2 November 17-December 14, 2015
· HGS station has been decommissioned. 

[bookmark: _MON_1387176949][bookmark: _MON_1388382035][bookmark: _MON_1388382145][bookmark: _MON_1388384576][bookmark: _MON_1388385083][bookmark: _MON_1388993005][bookmark: _MON_1389598124][bookmark: _MON_1390809671][bookmark: _MON_1390809790][bookmark: _MON_1391499119][bookmark: _MON_1391501949][bookmark: _MON_1392188921][bookmark: _MON_1392615265][bookmark: _MON_1393314853][bookmark: _MON_1393834790][bookmark: _MON_1394437595][bookmark: _MON_1394437598][bookmark: _MON_1394437644][bookmark: _MON_1395039355][bookmark: _MON_1395831044][bookmark: _MON_1396332066][bookmark: _MON_1396332146][bookmark: _MON_1396332643][bookmark: _MON_1396332666][bookmark: _MON_1396332684][bookmark: _MON_1396332702][bookmark: _MON_1396862017][bookmark: _MON_1397634356][bookmark: _MON_1397634698]Operational Discrepancies:
· SG1 pass at 311/15:59z lost one scene and pre-cal for another scene
· Assumed snow on radome
· SG1 pass at 312/16:35z lost most of one scene, only able to recover ALI for it
· CDS# 263290 - Snow on radome 




UPCOMING EVENTS
· Testing and validation of FD systems on Windows 7, ASIST systems on RedHat upgrade
· Continuation of Phase 1/2 FD, waiting on WALLOPS support 
· [bookmark: _GoBack]Moving on to Phase 3 for central file hub, deliveries
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Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

4-Nov-15 308 5 5 0 0 10

5-Nov-15 309 5 1 0 0 6

6-Nov-15 310 5 4 0 0 9

7-Nov-15 311 5 4 0 0 9

8-Nov-15 312 4 5 0 0 9

9-Nov-15 313 6 5 0 0 11

10-Nov-15 314 1 8 0 0 9

31 32 0 0 63 Weekly Totals
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		4-Nov-15		308		5		5		0		0		10

		5-Nov-15		309		5		1		0		0		6

		6-Nov-15		310		5		4		0		0		9

		7-Nov-15		311		5		4		0		0		9

		8-Nov-15		312		4		5		0		0		9

		9-Nov-15		313		6		5		0		0		11

		10-Nov-15		314		1		8		0		0		9

		Weekly Totals				31		32		0		0		63
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